**Author's response to reviews**

**Title:** Predictors of self-rated health. A 12-month prospective study of IT and media workers.

**Authors:**

Dan Hasson (dan.hasson@ki.se)
Bengt B Arnetz (bengt.arnetz@pubcare.uu.se)
Tores Theorell (tores.theorell@ipm.ki.se)
Ulla Maria Anderberg (ullamaria.anderberg@pubcare.uu.se)

**Version:** 7 **Date:** 19 July 2006

**Author's response to reviews:**

Dear Population Health Metrics Editorial Team,

Now the additional requested formatting changes have been done.

1. 'Title page' and 'Authors' have been removed from the Title page as well as the phone and fax number.

2. Author affiliations are now depicted with superscript numbers.

3. The acronym sub-section has been removed from the Abstract.

4. The repeated title on page 3 has been removed.

5. It is indicated that all authors have read and approved the final manuscript under the heading 'Authors' contributions'.

6. The figure legend has been inserted after the references and the section has been entitled 'Figure legend'. Furthermore, the next section is now entitled Tables.

With best wishes

Dan Hasson